Author's Response To Reviewer Comments



_____ _____ _____ Concerning software repository: _____ -----_____ Dear Editor, hereafter we reported general information about the current software repositories: 1. The rCASC package is available at this github repository: https://github.com/kendomaniac/rCASC 2. All the docker images are stored in the docker hub: docker.io/repbioinfo/ 3.GUI for rCASC is available at this github repository: https://github.com/mbeccuti/4SeqGUI 4. All the sample data are retrievable at 130.192.119.59 and the paths are indicated in the supplementary material. Moreover we registered rCASC in bio.tools and in SciCrunch.org (id: SCR_017005)

Clo<u>s</u>e